Author's response to reviews

Title: ACTN4 and the pathways related to cell motility and adhesion contribute to the brain metastasis of lung cancer

Authors:
Yufei Gao (gaoyufei1975@sina.cn)
Guanghu Li (ghl1001@163.com)
Liankun Sun (sunlk@jlu.edu.cn)
Yichun He (Heyichun1989@sina.com)
Xiaoyan Li (Lixiaoyan0510@163.com)
Zhi Sun (13514303555@163.com)
Jihan Wang (513837742@qq.com)
Yang Jiang (jy7555@163.com)
Jingwei Shi (Shi123jingwei@163.com)

Version: 2
Date: 25 November 2014

Author's response to reviews:

Dear Editor,

We would like to submit our manuscript entitled “ACTN4 and the pathways related to cell motility and adhesion contribute to the brain metastasis of lung cancer” for publication as an original article in BMC Cancer.

We agree to transfer our manuscript to BMC Cancer

This paper has not been published elsewhere in whole or in part. All authors have read and approved the content, and agree to submit it for consideration for publication in your journal. There are no ethical/legal conflicts involved in the article. This manuscript has been edited and proofread by Medjaden Bioscience Limited.

Your consideration for this manuscript is highly appreciated.

We look forward to hearing from you soon.

Yours sincerely,

*Corresponding author: Jun Wang
Jingwei Shi and Yang Jiang
Department of Laboratory Medicine Center and Department of Colorectal Surgery, China-Japan Union Hospital, Jilin University, Changchun 130033, China;
*Email: Jingwei Shi, shi123jingwei@163.com; Yang Jiang, jy7555@163.com.
Fax: +86-0431-88975348; Tel: +8618844199977